Author’s response to reviews

Title: The incorporation of focused history in Checklist for Early Recognition and Treatment of Acute Illness and Injury

Authors:
Namita Jayaprakash (Namita.Jayaprakash@gmail.com)
Rashid Ali (ali.rashid@mayo.edu)
Rahul Kashyap (kashyap.rahul@mayo.edu)
Courtney Bennett (bennett.courtney@mayo.edu)
Alexander Kogan (kogan.alexander@mayo.edu)
Ognjen Gajic (gajic.ognjen@mayo.edu)

Version: 2 Date: 08 Aug 2016

Author’s response to reviews:
To the editors of BMC emergency medicine,

This submission addresses the following minor revisions that were requested:

1. Please kindly indicate in the title page that who is the corresponding authors of this study.
2. Please kindly include all authors' emails on title page.
3. The individual contributions of authors to the manuscript should be specified in Authors' Contributions section.

The manuscript title page has been revised to include the corresponding author and all authors' email addresses. The individual contributions of each author is now also highlighted in the contributions section.
Regards,

N. Jayaprakash, R. Ali, R. Kashyap, C. Bennett, A. Kogen, O. Gajic